Author's response to reviews

Title: The impact of oral health conditions, socioeconomic status and use of specific substances on quality of life of addicted persons

Authors:

Tais C.N. Marques (tais.cnm@gmail.com)
Karin L.M. Sarracini (karin.migliato@gmail.com)
Karine L. Cortellazzi (karine.cortellazzi@gmail.com)
Fábio L. Mialhe (mialhe@fop.unicamp.br)
Marcelo C. Meneghim (meneghim@fop.unicamp.br)
Antonio C. Pereira (apereira@fop.unicamp.br)
Glaucia M.B. Ambrosano (ambrosano@fop.unicamp.br)

Version: 3 Date: 21 October 2014

Author's response to reviews: see over
Piracicaba, October 21, 2014.

To Editor-in-Chief,
BMC Oral Health,
Professor Magdalena Morawska

We are submitting the manuscript “The impact of oral health conditions, socioeconomic status and use of specific substances on quality of life of addicted persons” for your consideration for publication in Health and Quality of Life Outcomes. The authors assure that the manuscript is original, and it has not been published, partly or in its entirety, in any other journal, printed or electronic. The authors have no conflict of interest.

The corresponding author responsible for communication with the others authors of the manuscript is Tais Cristina Nascimento Marques:
Department of Community Dentistry
Piracicaba Dental School – University of Campinas.
Avenida Limeira, 901 – Piracicaba, SP – Brazil.
Zip Code: 13414-018. P.O. Box 52.
Phone: 55 (19) 2106 5209; Fax: 55 (19) 2106 5218.
The manuscript contains 2 Tables, and according to the authors' evaluations, they can be produced in black and white, without the need for colored versions.

Sincerely,

Tais Cristina Nascimento Marques

Karin Luciana Migliato Sarracini

Karine Laura Cortellazzi

Fábio Luiz Mialhe

Marcelo de Castro Meneghim

Antônio Carlos Pereira

Glaucia Maria Bovi Ambrosano